n Midterm Review n

2Chapters CoverednCh. 1nCh. 2nCh. 3nCh. 11nCh. 14nCh. 15

1Midterm ReviewIn Text: Chapters 1, 2, 3, 11, 14, 15

n Midterm Review n

4Ch. 2: History and EvolutionnWhat are the major contribution(s) of eachlanguage?nExamples:nSupport for recursionnIntroduction of block structurenFirst in a paradigmnFirst formally described syntaxnEtc.nMost important to review: FORTRAN, LISP,ALGOL

n Midterm Review n

3Ch. 1: IntroductionnWhy study PLs?nLanguage evaluation criterianTwo main influences on PL designn5 paradigmsnImplementation strategies

n Midterm Review n

6Ch. 11: Object-Oriented PLsnWhat are the three defining characteristicsof OOP languages?nWhat are the other hallmarks of OOP?nDefinition of OOP termsnFamiliarity with SmalltalknDesign issues for OOPLs

n Midterm Review n

5Ch. 3: Syntax and SemanticsnBNF & EBNFnWhat makes a grammar ambiguous?nHow do you disambiguate a grammar?nWhat does it mean for two grammars to�generate the same language�?nOperational semanticsnAxiomatic semanticsnApplying axiomatic semantics to codesegmentsnLoop invariants, 5 criterianDenotational semantics

n Midterm Review n

8Ch. 15: Logic ProgrammingnWhat are the hallmarks of LP?nDefinition of LP termsnProlog syntaxnYou should be able to read Prolog code

n Midterm Review n

7Ch. 14: Functional ProgrammingnWhat are the hallmarks of FP?nDefinition of FP termsnScheme syntaxnYou�ll have to write a Scheme function